Author’s response to reviews

Title: Palliative home parenteral nutrition in patients with ovarian cancer and malignant bowel obstruction: experiences of women and family caregivers.

Authors:

Anne Marie Sowerbutts (annemarie.sowerbutts@manchester.ac.uk)
Simon Lal (Simon.Lal@srf.nhs.uk)
Jana Sremanakova (jana.sremanakova@manchester.ac.uk)
Andrew Clamp (andrew.clamp@christie.nhs.uk)
Gordon Jayson (gordon.jayson2@christie.nhs.uk)
Antje Teubner (antje.teubner@srf.nhs.ukb)
Lisa Hardy (lisa.hardy@mft.nhs.uk)
Chris Todd (Chris.Todd@manchester.ac.uka)
Anne-Marie Raftery (AnneMarie.Raftery@christie.nhs.uk)
Eileen Sutton (E.Sutton@bristol.ac.uk)
Robert Morgan (Robert.Morgan@christie.nhs.uk)
Alexander Vickers (alexander.vickers@nhs.netc)
Sorrel Burden (Sorrel.Burden@manchester.ac.uk)

Version: 2 Date: 07 Dec 2019

Author’s response to reviews:

Dear Dr Lowe

Please find our response to your comments below:

1. Identifying information

We have removed all the ages of the women from the main paper and from additional file 1

2. Ethics approval and consent to participate
3. Authors’ contributions

RDM and AJV assisted with reviewing, revising and editing the paper. RDM also interpreted the data, and both provided input and interpretation of surgical and chemotherapy data. It was my mistake leaving them off these sections. In addition they have both approved the submitted version and to have agreed both to be personally accountable for the author’s own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Kind regards

Anne Marie Sowerbutts on behalf of the research team